



April 10, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 6684-revision.doc).

**Title:** Intra-abdominal esophageal duplication cyst: a case report and review of the literature.

**Author:** Petrus Sebastianus Simon Castelijns, Karlijn Woensdregt, Brigiet Hoevenaars, Gelde Arie Pieter Nieuwenhuijzen

**Name of Journal:** *World Journal of Gastrointestinal Surgery*

**ESPS Manuscript NO:** 6684

The manuscript has been improved according to the suggestions of reviewers:

1 The format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) **Reviewer 02540153:** We performed some language polishing and adjusted the abstract.

(2) **Reviewer 00068809:** This reviewer had no comments for the authors.

(3) **Journal Chief Editor:** We summarized each of the case characteristics, the clinical diagnosis, the differential diagnosis, the imaging diagnosis, and the pathological diagnosis in one sentence. Additionally we adjusted the comments on figure 1 and we adjusted the arrows and numbers in figure 4 to clarify the meaning of them.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastrointestinal Surgery*

Sincerely yours,

A handwritten signature in black ink, appearing to read 'Woensdregt', written over a long horizontal line.

Karlijn Woensdregt, MD

Department of Surgery

Catharina Hospital Eindhoven

Eindhoven, Michelangelolaan 2

The Netherlands

Fax: +31-40-2443370

E-mail: karlijn.woensdregt@catharinaziekenhuis.nl